Author’s response to reviews

Title: Antimicrobial and antiproliferative activity of essential oil, aqueous and ethanolic extracts of Ocimum micranthum Willd leaves

Authors:
Isabel Caamal-Herrera (isabel.caamal@cinvestav.mx)
Leydi Carrillo-Cocom (leydi.carrillo@correo.uady.mx)
Diana Escalante-Réndiz (derendiz@correo.uady.mx)
Diana Araíz-Hernández (diana_1157415@itesm.mx)
José Azamar (jose.azamar@cinvestav.mx)

Version: 3 Date: 31 Jan 2018

Author’s response to reviews:

Mérida, Yuc. January 31, 2018

Liam Messin

BMC Complementary and Alternative Medicine

Please find enclosed the corrections to the manuscript entitled "Antimicrobial and antiproliferative activity of essential oil, aqueous and ethanolic extracts of Ocimum micranthum Willd leaves" (BCAM-D-17-00261R2).

1. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Figure titles/legends were moved to section Figure Legends

Evidence page 24 line 589-600

Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

The Figure files contain only image/graphic with associated keys/annotations.

See the figure files attached.
2. Please include the email addresses for all authors on the title page. The corresponding author should still be indicated.

The email addresses for all authors were included on the title page
Evidence: page 1 line 17-21.

3. Please include the reference ID for the specimen that was deposited in the herbarium.

The reference ID for the specimen was included in the methods section
Evidence: page 6 line 133.

4. Please separate your Results and Discussion sections, as standard for articles in BMC Complementary & Alternative Medicine: https://bmccomplementalternmed.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article

Results and Discussions sections were separated
Evidence:
Results: page 11-14 line 264-347
Discussion: page 14-18 line 349-439

5. Please clarify in the Availability of data and materials section who or from where the data will be available from.

The availability of data and materials section was clarified
Evidence: page 19 line 463-464.

6. Please clarify the role that the Funding body had in this study in the Funding section.

The role of the Funding body was clarified in the Funding section
Evidence: Page 19 line 468-469.
7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

This point was attended.

Best regard,

Dr. José Antonio Azamar Barrios